Author's response to reviews

Title: Mammoth Interatrial Septal Aneurysm in the ICE Age

Authors:

Ravinay Bhindi (ravinay.bhindi@cardiov.ox.ac.uk)
John Timperley (john.timperley@orh.nhs.uk)
Oliver J Ormerod (Oliver.ormerod@orh.nhs.uk)

Version: 2 Date: 28 July 2007

Author's response to reviews:

Dear Dr Picano

Thank you for your email regarding our case report. We are pleased that Cardiovascular Ultrasound will accept our paper after minor changes which have now been incorporated.

Specific points

1. As suggested the title has been changed to 'Mammoth interatrial septal aneurysm in the ICE age'.

2. The abstract is now presented separately from the text

3. The reference has been reformatted as specified by the journal

Thank you again for your prompt review of the manuscript.

Dr Ravinay Bhindi